
\documentclass[twoside,draft,reqno]{birkart}

\usepackage{amssymb}

%%\nofiles
%%\writetoctrue

%%\input defs

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{claim}[theorem]{Claim}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{problem}[theorem]{Problem}

\newenvironment{Proof}{\removelastskip\par\medskip
\noindent{\em Proof.} \rm}{\penalty-20\null\hfill$\square$\par\medbreak}

\newenvironment{Proofx}{\removelastskip\par\medskip
\noindent{\em Proof.} \rm}{\par}

\begin{document}
\begin{thebibliography}{99}

\itemsep=\smallskipamount

\bibitem{A} S. Abhyankar, {\em Local rings of high embedding dimension,}
Amer. J. Math., {\bf 89} (1967), 1073--1077.

\bibitem{A1} J. Alexander, {\em Singularit\'{e}s imposable en position general
\'{a} une hypersurface projective,} Compositio Math, {\bf 68} (1988), 305--354.

\bibitem{AH1} J. Alexander and A. Hirschowitz, {\em La m\'{e}thode d'Horace
\'{e}clat\'{e}: Application \'{a} l'in\-ter\-po\-la\-tion en degr\'{e}
quatre,} Inv. Math, {\bf 107} (1992), 585--602.

\bibitem{AH2} J. Alexander and A. Hirschowitz, {\em Un Lemme d'Horace
diff\'{e}rentiel: Application aux singularit\'{e}s hyperquartiques de $\bf P^5$,}
J. Alge. Geo. {\bf 1} (1992), 411--426.

\bibitem{An} D. Anick, {\em Thin algebras of embedding dimension three,}
J. Algebra, {\bf 100} (1986), 235--259.

\bibitem{BS} D. Bayer and M. Stillman, {\em A criterion for detecting {\it
m}-regularity,} Inv. Math, {\bf 87} (1987), 1--11.

\bibitem{BI} D. Bernstein and A. Iarrobino, {\em A nonunimodal graded Gorenstein
Artin algebra in codimension five,} Comm. Algebra, {\bf 20} (1992), 2323--2336.

\bibitem{BGM} A. M. Bigatti, A. V. Geramita and J. Migliore, {\em Geometric
consequences of extremal behaviour in a Theorem of Macaulay,} Trans.
A.M.S., {\bf 346 (1)} (1994), 203--235.

\bibitem{BL} M. Boij and D. Laksov, {\em Nonunimodality of graded Gorenstein
Artin algebras,} Proc. A. M. S., {\bf 120} (1994), 1083--1092.

\bibitem{CG} M. V. Catalisano and A. Gimigliano, {\em On the Hilbert function of
fat points on a rational normal cubic,} Preprint,(1996)

\bibitem{CTV} M. V. Catalisano, N. V. Trung and G. Valla {\em A sharp bound for
the regularity index of fat points in general position,} Proc. A. M. S., {\bf 118}
(1993), 717--724.

\bibitem{Ch} K. Chandler, {\em Hilbert functions of dots in linear general
position,} Zero-dimensional Schemes, Proceedings of the International
Conference held in Ravello, June 8--13, 1992, Edited by F. Orecchia and L. Chiantini,
Walter De Gruyter (Berlin-New York)
{\bf 1994}, 65--79.

\bibitem{DGO} E. D. Davis, A. V. Geramita and F. Orecchia, {\em Gorenstein
Algebras and the Cayley-Bacharach theorem,} Proc. A. M. S., {\bf 93} (1985),
593--598.

\bibitem{DV} E. De Negri and G. Valla,
{\em The $h$-vector of a Gorenstein codimension three domain,} Nagoya Math. J.,
{\bf 138} (1995), 113--140.

\bibitem{ER} R. Ehrenborg and G. C. Rota,
{\em Apolarity and canonical forms for homogeneous polynomials,} Europ. J. Comb.,
{\bf 14} (1993), 157--182.

\bibitem{Ei} D. Eisenbud, {\em Commutative Algebra with a view toward Algebraic
Geometry,} Graduate Texts in Mathematics, {\bf 150} {1995}, Springer-Verlag,
New York.

\bibitem{EGH} D. Eisenbud, M. Green and J. Harris, {\em Higher Castelnuovo
Theory,} Asterisque, {\bf 218} (1993), 187--202.

\bibitem{E} J. Elias, {\em Characterization of the Hilbert-Samuel polynomial of
curve singularities,} Comp. Math, {\bf 74} (1990), 135--155.

\bibitem{E2} J. Elias, {\em The conjecture of Sally on the Hilbert function for
curve singularities,} J. of Algebra,
{\bf 160} (1993), 42--49.

\bibitem{EV} J. Elias and G. Valla, {\em Rigid Hilbert Functions,} J. Pure and
Appl. Algebra, {\bf 71} (1991), 19--41.

\bibitem{ERV} Elias, J., Rossi, M.E., Valla, G., {\em On the coefficients of the
Hilbert polynomial,}
J. Pure Appl. Algebra., {\bf 108} (1996), 35--60.

\bibitem{EI} J. Emsalem and A. Iarrobino, {\em Inverse System of a symbolic
power, I,} J. of Algebra, {\bf 174} (1995), 1080--1090.

\bibitem{FP} C. Fefferman and D. Phong, {\em The uncertainty principle and sharp
Garding inequalities,} Comm. Pure Appl.Math, {\bf 34} (1981), 285--331.

\bibitem{F} R. Fr\"oberg, {\em An inequality for Hilbert series of graded
algebras,} Math. Scand., {\bf 56} (1985), 117--144.

\bibitem{tony} A. V. Geramita, {\em Inverse Systems of fat points: Waring's
Problem, Secant Variety of Veronese Varieties and Parameter Spaces for
Gorenstein Ideals,} Queen's Papers Pure App. Math., Vol. X, The Curves Seminar
at Queen's, {\bf 102} (1996), 1--104.

\bibitem{GGR} A. V. Geramita, D. Gregory and L. Roberts, {\em
Monomial ideals and points in projective space,} J. Pure and Appl. Alg., {\bf
40 } (1986),33--62

\bibitem{GKR} A. V. Geramita, M. Kreuzer and L. Robbiano, {\em
Cayley-Bacharach schemes and their canonical modules,} Trans. A. M. S. , {\bf 339 }
(1993),163--189.

\bibitem{GM} A. V. Geramita and J. C. Migliore, {\em Hyperplane sections of a
smooth curve in ${\bf P}^3$,} Comm. in Alg., {\bf 17 } (1989), 3129--3164.

\bibitem{GP} L. Gruson and C. Peskine, {\em Genre des courbes de l'espace
projectif,} Algebraic Geometry, Lect. Notes Math., {\bf 687} (1978) Springer.

\bibitem{GR} S. K. Gupta and L.G. Roberts, {\em Cartesian squares and ordinary
singularities of curves,} Comm. in Alg., {\bf 11(1) } (1983), 127--182.

\bibitem{Ha} J. Harris, {\em Curves in projective space,}
Montreal: Les Presses de l'Universite' de Montreal, {\bf 1982}.

\bibitem{Ha1} J. Harris, {\em Algebraic Geometry. A first course,}
Graduate Texts in Mathematics, {\bf 133} (1992), Springer-Verlag, New York.

\bibitem{Hart} R. Hartshorne , {\em Connectedness of the Hilbert scheme,}
Math. Inst. des Hautes Etudes Sci., {\bf 29} (1966), 261--304.

\bibitem{HTV} J. Herzog, N. V. Trung and G. Valla, {\em On hyperplane sections
of reduced irreducible varieties of low codimension,} J. Math. Kyoto Univ, {\bf
34} (1994), 47--72.

\bibitem{Hi} T. Hibi, {\em Flawless $O$-sequences and Hilbert functions of
Cohen-Macaulay integral domains,} J. Pure and Appl. Algebra, {\bf 60} (1989),
245--251.

\bibitem{Hil} D. Hilbert, {\em \"Uber die Theorie der algebraischen Formen,}
Math. Ann., {\bf 36} (1890), 473--534.

\bibitem{H} A. Hirschowitz, {\em La m\'{e}thode d'Horace pour l'interpolation
\'{a} plusieurs variables,} Manus. Math, {\bf 50} (1985), 337--388.

\bibitem{HL} M. Hochster and D. Laksov, {\em The linear syzygies of generic
forms,} Comm. Algebra, {\bf 15} (1987), 227--239.

\bibitem{HM} S. Huckaba and T. Marley, {\em Hilbert coefficients and the depths
of associated graded rings,} Preprint, (1996).

\bibitem{Hun} C. Huneke, {\em Hilbert functions and symbolic powers,}
Michigan Math. J., {\bf 34} (1987), 293--318.

\bibitem{K} K. Kubota, {\em On the Hilbert-Samuel function,}
Tokyo J. Math., {\bf 8} (1985), 439--448.

\bibitem{I} A. Iarrobino, {\em Inverse System of a symbolic
power, II. The Waring Problem for forms,} J. of Algebra, {\bf 174} (1995),
1091--1110.

\bibitem{M1} F. S. Macaulay, {\em Algebraic theory of modular systems,}
Cambridge Tracts {\bf 16,} Cambridge University Press, Cambridge, U. K., {1916.}

\bibitem{M2} F. S. Macaulay, {\em Some properties of enumeration in the theory
of modular systems,} Proc. London Math. Soc., {\bf 26} (1927), 531--555.

\bibitem{MR} R. Maggioni and A. Ragusa, {\em The Hilbert Function of generic
plane sections of curves in ${\bf P}^3,$} Inv. Math., {\bf 91} (1988),
253--258.

\bibitem{M} T. Marley, {\em The coefficients of the Hilbert polynomial and the
reduction number of an ideal,} J. London Math. Soc., {\bf 40} (1989), 1--8.

\bibitem{MP} D. McDuff and L. Polterovich, {\em Symplectic packings and
Algebraic Geometry,} Inv. Math., {\bf 115} (1994), 405--429.

\bibitem{Na} M. Nagata, {\em On the 14-th problem of Hilbert,} Am. J. Math., {\bf
81} (1959), 766--772.

\bibitem{NA} M. Narita, {\em A note on the coefficients of Hilbert
characteristic functions in semi-regular local rings,} Proc. Cambridge Philos.
Soc., {\bf 59} (1963), 269--275.

\bibitem{NR} G. Niesi and L. Robbiano, {\em Disproving Hibi's conjecture with
CoCoA or projective curves with bad Hilbert functions,} in: F. Eyssette and A.
Galligo, Eds., Proceedings of Mega 92 (Birkh\"auser, Boston) (1993), 195--201.

\bibitem{N} D. G. Northcott, {\em A note on the coefficients of the abstract
Hilbert Function,} J. London Math. Soc., {\bf 35} (1960), 209--214.

\bibitem{O} A. Ooishi, {\em $\Delta$-genera and sectional genera of commutative
rings,} Hiroshima Math. J., {\bf 17} (1987), 361--372.

\bibitem{Or} F. Orecchia, {\em One-dimensional local rings with reduced
associated graded ring and their Hilbert functions,} Manus. Math., {\bf 32}
(1980), 391--405.

\bibitem{RR} L. J. Ratliff and D. Rush, {\em Two notes on reductions of ideals,}
Indiana Univ. Math. J, {\bf 27} (1978), 929--934.

\bibitem{RV} M. E. Rossi and G. Valla, {\em Multiplicity and $t$-isomultiple
ideals,} Nagoya Math. J., {\bf 110} (1988), 81--111.

\bibitem{RV2} M. E. Rossi and G. Valla, {\em A conjecture of J. Sally,} Comm. in
Algebra, {\bf 24(13)} (1996), 4249--4261.

\bibitem{S0} J. Sally, {\em On the associated graded ring of a local
Cohen-Macaulay ring,} J. Math Kyoto Univ, {\bf 17} (1977), 19--21.

\bibitem{S4} J. Sally, {\em Super regular sequences,} Pac. J. of Math , {\bf 84}
(1979), 475--481.

\bibitem{SG} J. Sally, {\em Good embedding dimension for Gorenstein
singularities,} Math. Ann., {\bf 249} (1980), 95--106.

\bibitem{S1} J. Sally, {\em Cohen-Macaulay local ring of embedding dimension
$e+d-2$,} J. Algebra, {\bf 83} (1983), 325--333.

\bibitem{S3} J. Sally, {\em Hilbert coefficients and reduction number 2,} J.
Algebraic Geometry, {\bf 1} (1992), 325--333.

\bibitem{S2} J. Sally, {\em Ideals whose Hilbert function and Hilbert polynomial
agree at $n=1$,} J. Algebra, {\bf 157} (1993), 534--547.

\bibitem{Singh} B. Singh {\em Effect of a permissible blowing-up on the
local Hilbert function,} Inv. Math., {\bf 26} (1974), 201--212.

\bibitem{St1} R. Stanley, {\em Hilbert functions of graded algebras,} Adv. in
Math, {\bf 28} (1978), 57--83.

\bibitem{St2} R. Stanley, {\em On the Hilbert function of a graded
Cohen-Macaulay domain,} J. Pure Appl. Algebra, {\bf 73} (1991), 307--314.

\bibitem{St3} R. Stanley, {\em Combinatorics and Commutative Algebra,}
Birkh\"auser, Boston, {\bf 1983}.

\bibitem{T} A. Terracini, {\em Sulla rappresenatzione di coppie di forme
ternarie mediante somme di potenze di forme lineari,} Ann. Mat. Pura Appl. Ser.
(3), {\bf 24} (1915), 1--10.

\bibitem{TV} N. V. Trung and G. Valla, {\em On zero-dimensional subschemes of a
complete intersection,} Math. Z, {\bf 219} (1995), 187--201.

\bibitem{VV} P. Valabrega and G. Valla, {\em Form rings and regular sequences,}
Nagoya Math. J., {\bf 72} (1978), 93--101.

\bibitem{V} G. Valla, {\em On the Betti numbers of perfect ideals,} Comp. Math.,
{\bf 91} (1994), 305--319.

\end{thebibliography}

\setcounter{page}{293}

\title[Sample Article Title]
{Sample Article Title to Illustrate\\
the Use of Birkart.cls}
\author[J.~Author]{James Author}

\address{%%
Department of Mathematics,\br
University of Sampleville,\br
35, Gau\ss\ Avenue\br
U-314159 Sampleville, Utopia}

\email{author@math.unisamp.ut}

\begin{abstract}
Since a projective
variety
$V=\mathcal{Z}(I)\subseteq {\bf P}^n$ is an
intersection of hypersurfaces, one of the most basic problems we can pose in
relation to $V$ is to describe the hypersurfaces containing it. In
particular, one would like to know the maximal number of linearly independent
hypersurfaces of each degree containing $V,$ that is to know the dimension of
$I_d,$ the vector space of homogeneous polynomials of degree
$d$ vanishing on $V$ for various $d.$
\end{abstract}

\maketitle

\section{Introduction}
Since one knows the dimension
$\binom{n+d}{d}$ of the space of all forms of degree
$d$, knowing the dimension of $I_d$ is
equivalent to knowing the {\bf Hilbert function} of the homogeneous
coordinate ring $A=k[X_0,\dots,X_n]/I$ of $V$, which is the vector
space dimension of the degree
$d$ part of $A.$

In his famous paper ``\"Uber die Theorie der algebraischen Formen'' (see \cite
{Hil}) published a century ago, Hilbert proved that a graded module $M$ over a
polynomial ring has a finite graded free resolution, and concluded from this
fact that its Hilbert function is of polynomial type. The {\bf Hilbert
polynomial} of a graded module is thus the polynomial which agrees with the Hilbert
function $H_M(s)$ for all large $s.$ Hilbert's insight was that all the
information encoded in the infinitely many values of the Hilbert function can be
read off from just finitely many of its values.

\subsection{First subsection}
Hilbert's
original motivation for studying these numbers came from invariant theory. Given
the action of a group on the linear forms of a polynomial ring, he wanted to
understand how the dimension of the space of invariant forms of degree $d$ can
vary with $d.$

\subsubsection{First subsubsection}
The Hilbert function of the homogeneous coordinate
ring of a projective variety $V,$ which classically was called the postulation of
$V,$ is a rich source of discrete invariants of
$V$ and its embedding. The dimension, the degree and the arithmetic genus of $V$
can be immediately computed from the generating function of the Hilbert function
of $V$ or from its Hilbert polynomial.

As for the Hilbert polynomial, there are two
important geometric contexts in which it appears. The first is the
Riemann-Roch theorem which plays an enormously important role in Algebraic
Geometry. This celebrated formula arises
from the computation of a suitable Hilbert polynomial. Secondly, the
information contained in the coefficients of the Hilbert polynomial is usually
presented in Algebraic Geometry by giving the Chern classes of the corresponding
sheaf, a set of different integers, which can be deduced from the coefficients,
and from which the coefficients can also be deduced (see \cite {Ei}, pg. 44).

\subsubsection{Second subsubsection}
The
influence of Hilbert's paper on Commutative Algebra has been tremendous. Both
free resolutions and Hilbert functions have fascinated mathematicians for a
long time. In spite of that, many central problems still remain open. The last
few decades have witnessed more intense interest in these objects. I believe this
is due to two facts. One is the arrival of computers and their breathtaking
development. The new power in computation has prompted a
renewed interest in the problem of effective construction in Algebra. The ability
to compute efficiently with polynomial equations has made it possible to
investigate complicated examples that would be impossible to do by hand and has
given the right feeling to tackle more difficult questions.

The second is the work of Stanley in connection with Algebraic
Combinatorics. In 1978 R. Stanley published the fundamental paper
``Hilbert
functions of graded algebras'' (see \cite {St1}) where he related the
study of the
Hilbert function of standard graded algebras to several basic problems in
Combinatorics. An introduction to this aspect of Commutative Algebra is given in
Stanley's monograph \cite {St3} which is well known as the ``green
book''.

From the point of view of the theory of Hilbert functions, one of
the greatest merits of Stanley's work was to restate and explain, in the right
setting, a fundamental theorem of Macaulay which characterizes the Hilbert
functions of homogeneous $k$-algebras. This theorem is a great source of
inspiration for many researchers in the field.

All the above arguments suffice to
justify the choice of ``Hilbert functions'' as the right topic for a postgraduate
course in Commutative Algebra. I would however add some further motivation. One
of the delightful things about this subject is that one can begin studying it in
an elementary way and, all of a sudden, one can front extremely challenging and
interesting problems. An
example will help to clarify this. The possible Hilbert functions of a
homogeneous Cohen-Macaulay algebra are easily characterized by using
Macaulay's theorem and a standard graded prime avoidance theorem. The
corresponding problem for
a Cohen-Macaulay domain is completely open and we do not even have a
guess as to the possible structure of the corresponding numerical functions.

\subsection{Second subsection}
Further, following what is written at the beginning of Eisenbud's
excellent book \cite{Ei}, ``It has seemed to me for a long time that Commutative
Algebra is best practiced with knowledge of the geometric ideas that played a
great role in its formation: in short, with a view toward Algebraic Geometry''. And
what better topic than Hilbert functions to give a concrete example of this
way of thinking?

Since I had
to choose among the many different scenes which compose the picture, I was very
much influenced in my choice by what I know better and what I have worked on
recently. This means that the chapters I present here are by no means the most
important in the theory of Hilbert functions. However I was also guided by the
possibility of inserting open problems and conjectures more than results.

Besides the basic definitions,
the first two sections are devoted to the problem of characterizing the possible
numerical functions which are the Hilbert functions of some graded algebras with
special properties e.g. reduced, Cohen-Macaulay or Gorenstein. Since the
main problems arise when the algebras are integral domains, we will show that, by
the classical Bertini theorem, the
$h$-vector of a Cohen-Macaulay graded domain of dimension bigger than or equal
to two, is the same as that of the homogeneous coordinate ring of an
arithmetically Cohen-Macaulay projective curve in a suitable projective space.
But, by a more recent result of J. Harris on the generic hyperplane section of a
projective curve, this
$h$-vector is also the
$h$-vector of the homogeneous coordinate ring of a set of points in Uniform
Position. This gives a very interesting shift from a purely algebraic approach to
a more geometric context which has been very useful.

In the
third section we introduce a list of conjectures recently made by Eisenbud, Green
and Harris and which are closely related to the topics introduced in the first
two sections. These
conjectures mainly deal with the problem of finding precise
bounds on the multiplicity of special Artinian graded algebras, but they can also
be read in a more geometric contest. A stronger form
of some of these conjectures is a guess which extends, in a very natural way, the
main theorem of Macaulay we have introduced above. Despite the fact that
we have almost no answer to these questions, they fit very well into the picture,
because they are so easy to formulate and so difficult to solve.

A short fourth section is devoted to a longstanding question on
the possible Hilbert function of generic graded algebras. A solution in the case
of a polynomial ring in two variables is given which uses a nice argument related
to the Gr\"obner basis theory of ideals in the polynomial ring. A very natural
problem in this theory closes the section.

In the fifth section
we discuss some problems related to the Hilbert function of a scheme of fat
points in projective space. We will try to explain how the knowledge of the
postulation of these zero-dimensional schemes can be used to study the Waring
problem for forms in a polynomial ring and the symplectic packing problem for
the four-dimensional sphere.

As for the first problem,
following the approach used by T. Iarrobino in his recent work, we will show how a
deep theorem of Alexander and Hirschowitz on the Hilbert function of the scheme
of generic double fat points gives a complete solution to the Waring problem
for forms, which is the old problem of determining the least integer $G(j)$
such that the generic form of degree $j$ in $k[X_0,\dots,X_n]$ is the sum of $G(j)$
powers of linear forms.

As for the second problem, we just
present the relationship between an old conjecture of Nagata on the postulation of
a set of fat points in ${\bf P}^2$ and the problem on the existence of a full
symplectic packing of the four dimensional sphere, as presented in the work of
McDuff and Polterovich.

In the last section we come to the
Hilbert function of a local Cohen-Macaulay ring and present several results and
conjectures on this difficult topic. Since the associated graded ring of a
local Cohen-Macaulay ring can be very bad (there do exist local complete
intersection domains whose associated graded ring has depth zero), very little is
known on the possible Hilbert functions of this kind of ring, even if this has
strong connections with the well developed theory of singularities.

Starting from the classical results of S. Abhyankar, D. Northcott and J.Sally, we
discuss how Cohen-Macaulay local rings which are extremal with
respect to natural numerical constraints on some of their Hilbert coefficients,
have good associated graded rings and special Hilbert functions. A recent result of
Rossi and Valla, which gives a solution to a longstanding conjecture made by
Sally, is also discussed at the end of the section.

\medskip

I am
personally grateful to the organizers of the school for giving me the possibility
to teach on my favourite topic. A number of people helped me a great deal in the
development of this manuscript. In particular, the section on Waring's problem is
very much influenced by a series of talks Tony Geramita gave in Genova last year
(see \cite{tony}). The last section on the Hilbert function in the local case grew
out of a long time cooperation with M. E. Rossi.

Finally I apologize to those whose work I may have failed to cite
properly. My feelings are best described by the following sentence which I found
in
\cite {Ha1}:

``Certainly, the absence of a reference for any particular discussion
should be taken simply as an indication of my ignorance in this regard, rather
than as a claim of originality.''

\section {Second Section}
Our standard assumption will be that $k$ is a field of characteristic zero, $R$
is the polynomial ring
$k[X_1,\dots,X_n]$ and $M$ a finitely generated graded
$R$-module such that $M_i=0$ if
$i<0.$ If $M$ is such a graded $R$-module, the homogeneous components
$M_n$ of $M$ are $k$-vector spaces of finite dimension.

\begin{definition} Let $M$ be a finitely generated graded
$R$-module. The numerical function
$$
H_M:{\bf N}\to {\bf N}
$$
defined as
$$
H_M(t)=dim_k(M_t)
$$
for all $t\in {\bf N}$ is the {\it Hilbert function}
of
$M.$ 

The power series
$$
P_M(z)=\sum_{t\in {\bf N}}H_M(t)z^t
$$
is called the {\it Hilbert Series} of
$M.$
\end{definition}

For example, for every $t\ge 0$ we have
$$
H_R(t)={\binom{n+t-1}{t}}
\qquad\mbox{and}\qquad
P_R(z)=\frac{1}{(1-z)^n}.
$$
The most relevant
property of the Hilbert series is the fact that it is additive on
short exact sequences of finitely generated $R$-modules. A classical result of
Hilbert says that the series
$P_M(z)$ is rational and even more,

\begin{theorem}\label{Hilbert-Serre}{\bf (Hilbert-Serre)} Let $M$
be a finitely generated graded $R$-module. Then there exists a polynomial
$f(z)\in {\bf Z}[z]$ such that
$$
P_M(z)=\frac{f(z)}{(1-z)^n}.
$$
\end{theorem}

It is easy to see that if $M\not= 0$ then the multiplicity of 1 as a root of
$f(z)$ is less than or equal to $n$ so that we can find a unique polynomial
$$
h(z)=h_0+h_1z+\cdots+h_sz^s\in {\bf Z}[z]
$$
such that
$h(1)\not = 0$ and for some integer $d,$ $0\le d\le n$
$$
P_M(z)=\frac{h(z)}{(1-z)^d}.
$$
The polynomial $h(z)$ is called {\it the
h-polynomial } of $M$ and the vector
$(h_0,h_1,\dots,h_s)$ {\it the h-vector} of $M.$

The integer $d$ is the
{\it Krull dimension} of $M.$ 

Now for every $i\ge 0,$ let
$$
e_i:=\frac{h^{(i)}(1)}{i!}
$$
and
$$
{\binom{X+i}{i}}:=\frac {(X+i)\cdots (X+1)}{i!}.
$$
Then it is easy to see that
the polynomial
$$
p_M(X):=\sum_{i=0}^{d-1}(-1)^ie_i{\binom{X+d-i-1}{d-i-1}}
$$
has rational
coefficients and degree $d-1;$
further for every $t>>0$ 
$$
p_M(t)=H_M(t).
$$
The polynomial $p_M(X)$ is called the
{\it Hilbert polynomial} of $M$ and its leading coefficient is
$$
\frac {h(1)}{(d-1)!}.
$$
This implies that
$e_0(M):=h(1)$ is a positive integer which is usually denoted simply by $e(M)$
and called {\it the multiplicity} of $M.$

If $d=0$ we define
$e(M)=dim_k(M).$

Another relevant property of Hilbert series is the so called
{\it sensitivity to regular sequences.} We recall that a sequence $F_1\dots,F_r$ of
elements of the polynomial ring $R$ is a {\it regular sequence} on a
finitely generated graded $R$-module $M$ if $F_1,\dots,F_r$ have positive degrees
and
$F_i$ is not a zero-divisor on $M$ modulo $(F_1,\dots,F_{i-1})M$ for
$i=1,\dots,r.$

If
$J$ is the ideal generated by the homogeneous polynomials $F_1,\dots,F_r,$ of
degrees
$d_1,\dots,d_r,$ one can prove that
$$
P_M(z)\le \frac{P_{M/JM}(z)}{\prod_{i=1}^r(1-z^{d_i})}
$$
with equality holding if
and only if the elements $F_1,\dots,F_r$ form a regular sequence on $M.$ This means
that if $L\in R_1$ is regular on $M,$ we have 
$$
P_{M/LM}(z)=P_M(z)(1-z).
$$
So
that the Hilbert function of the module $M/LM$ is given by the so called
{\it first difference function}
$\Delta H_M$ of $H_M$ which is defined by the formula
$$
\Delta H_M(t)=\left \{\begin{array}{ll}1 & \mbox{if $t=0$}\\H_M(t)-H_M(t-1) &
\mbox{if $t\ge 1.$ }\end{array}\right.
$$

Now we present a fundamental theorem, due to Macaulay (see \cite{M2}), describing
exactly those numerical functions which occur as the Hilbert function
$H_A(t)$ of a standard homogeneous
$k$-algebra
$A.$ Macaulay's theorem says that for each $t$ there is an upper bound for
$H_A(t+1)$ in terms of
$H_A(t)$, and this bound is sharp in the sense that any numerical function
satisfying it can be realized as the Hilbert function of a suitable
homogeneous
$k$-algebra.

Let $d$ be a positive integer. One can easily see
that any integer
$a$ can be written uniquely in the form
$$
a={\binom{k(d)}{d}}+{\binom{k(d-1)}{d-1}}+\cdots +{\binom{k(j)}{j}}
$$
where
$$
k(d)>k(d-1)>\cdots >k(j)\ge j\ge 1.
$$
For example, if
$a=49,$ $d=4,$ we get
$$
49={\binom{7}{4}}+{\binom{5}{3}}+{\binom{3}{2}}+{\binom{1}{1}}.
$$
Given
the integers $a$ and $d$, we let
$$
a^{<d>}:={\binom{k(d)+1}{d+1}}+{\binom{k(d-1)+1}{d}}+\cdots
+{\binom{k(j)+1}{j+1}}.
$$
Hence, for example,
$$
49^{<4>}={\binom{8}{5}}+{\binom{6}{4}}+{\binom{4}{3}}+{\binom{2}{2}}.
$$

\begin{theorem}\label{Macaulay theorem}{\bf (Macaulay)} Let $H:{\bf
N}\to
{\bf N}$ be a
numerical function. There exists a standard homogeneous $k$-algebra $A$ with
Hilbert function
$H_A=H$ if and only if
$H(0)=1$ and $H(t+1)\le H(t)^{<t>}$ for every $t\ge 1.$
\end{theorem}

A numerical function verifying the conditions of the above theorem is called an
{\it admissible} numerical function.

The following example demonstrates the effectiveness of Macaulay's theorem.
Let us check that
$1+3z+4z^2+5z^3+7z^4$ is not the Hilbert series of a homogeneous $k$-algebra. We
have 
$$
5={\binom{4}{3}}+{\binom{2}{2}}
$$
$$
5^{<3>}={\binom{5}{4}}+{\binom{3}
{3}}=6.
$$

In the same paper Macaulay produced an algorithm to construct,
given an admissible numerical function $H,$ an homogeneous $k$-algebra having it as
Hilbert function. 

Let $n=H(1);$ we fix in the set of monomials of
$R=k[X_1,\cdots,X_n]$ a total order compatible with the semigroup structure of this
set. Let us fix for example the {\it degree lexicographic order.} This is
the order given by
$$
X_1^{a_1}X_2^{a_2}\cdots X_n^{a_n}>X_1^{b_1}X_2^{b_2}\cdots X_n^{b_n}
$$
if and
only if either $\sum a_i>\sum b_i$ or $\sum a_i=\sum b_i$ and for some integer
$j<n$ we have
$a_1=b_1,\cdots, a_j=b_j,a_{j+1}>b_{j+1}.$

Macaulay proved that if for every $t\ge 0$ we delete the smallest
$H(t)$ monomials of degree
$t,$ the remaining monomials generate an ideal $I$ such that
$H_{R/I}(t)=H(t)$ for every $t\ge 0.$ 

The difficult part of the
proof is to show that, due to the upper bound
$H(t+1)\le H(t)^{<t>},$ if a monomial $M$ is in $I_t$, which means that it has not
been erased at level $t,$ then $MX_n$ is not erased at level $t+1$\ so that
$MX_1,\dots,MX_n\in I$ as we need.

For example the ideal $I$ in
$R=k[X_1,X_2,X_3]$ such that
$$
P_{R/I}(z)=\frac{1+z-2z^4+z^5}{(1-z)^2}
$$
and constructed by this algorithm, is
the ideal
$$
I=(X_1^2,X_1X_2^3,X_1X_2^2X_3).
$$

The ideal constructed following this
method is called a {\it lex-segment } ideal, in the sense that a $k$-basis of its
homogeneous part of degree $t$ is an initial segment of monomials in the given
order. Since it is clearly uniquely determined by the given admissible function,
it is called
{\it the lex-segment ideal} associated to the admissible numerical function.

This ideal has some very interesting extremal properties. For example it has the
biggest Betti numbers among the perfect ideals with the same multiplicity and
codimension (see
\cite{V}).

Macaulay's theorem is valid for any homogeneous $k$-algebra. It is not surprising
that additional properties yield further constraints on the Hilbert function. We
will discuss this feature for reduced, Cohen-Macaulay, Gorenstein and domain
properties. We start with the reduced case.

\begin{theorem}\label{reduced} Let $H:{\bf N}\to {\bf N}$ be a numerical
function. There exists a reduced homogeneous $k$-algebra $A$ with Hilbert function
$H_A=H$ if and only if either $H=\{1,0,0\dots\}$ or $\Delta H$ is admissible.
\end{theorem}
\begin{Proof}\ If $A=R/I$ is reduced it is clear that either $depth(A)>0,$ or
$I=(X_1,\dots,X_n).$ In the second case all is clear; in the first one we can find
a linear form $L$ which is a regular element on $A.$ Hence the difference of the
Hilbert function of $A$ is the Hilbert function of the graded algebra $A/LA.$
Conversely, if we are given a function whose difference is admissible, we can
construct the lex-segment ideal $J$ in the polynomial ring $S$ such that
$H_{S/J}=\Delta H.$ This is a monomial ideal which can be deformed to a radical
ideal by a general construction due to Hartshorne (see \cite{Hart} and \cite
{GGR}). This can be achieved in the following way. If
$m=X_1^{a_1}X_2^{a_2}\cdots X_n^{a_n}$ is a monomial, we introduce a new variable
$X_0$ and set
$$
l(m):=\prod_{j=1}^n\prod_{p=0}^{a_j-1}(X_j-pX_0).
$$
If $J=(m_1,\dots,m_s)$ is an
ideal generated by monomials, the ideal
$$
l(J):=(l(m_1),\dots,l(m_s))
$$
is a radical ideal in the polynomial ring
$R=k[X_0,\dots,X_n],$ such that $X_0$ is a regular element on $R/l(J)$ and
$$
S/J\simeq (R/l(J))/X_0(R/l(J)).
$$
Thus if $I\subseteq R$ is such a radical
deformation of
$J,$ we have
$$
P_{S/J}(z)=P_{(R/I)/X_0(R/I)}(z)=P_{R/I}(z)(1-z)
$$
so that $H_{S/J}=\Delta
H_{R/I}.$ Since $H_{S/J}=\Delta H=\Delta H_{R/I}$ we get $H=H_{R/I}.$
\end{Proof}

For example the numerical function $H=\{1,2,1,1,1,\dots\}$ is admissible
so it is the Hilbert function of a suitable graded $k$-algebra, but it is not the
Hilbert function of a reduced $k$-algebra since its difference function
$\{1,1,-1,0,0,\dots\}$ is not admissible.

We pass now to the
Cohen-Macaulay case and obtain the following characterization of the Hilbert
Series of Cohen-Macaulay homogeneous algebras.

\begin{theorem}\label{CM} Let $h_0,\dots,h_s$ be a finite sequence of positive
integers. There exists an integer
$d$ and a Cohen-Macaulay homogeneous
$k$-algebra $A$ of dimension $d$ such that
$$
P_A(z)=\frac{h_0+h_1z+\cdots +h_sz^s}{(1-z)^d}
$$
if and only if
$(h_0,\dots,h_s)$
is admissible.
\end{theorem}

\begin{Proof} If
$R/I$ is Cohen-Macaulay, by prime avoidance in the graded case, we can find a
maximal regular sequence of linear forms
$L_1,\dots,L_d$ such that $R/I$ and its Artinian reduction
$(R/I)/(L_1,\dots,L_d)(R/I)$ share the same
$h$-polynomial.
\end{Proof}

In particular the $h$-vector of a Cohen-Macaulay graded algebra has
positive coordinates.





\end{document}
